Author's response to reviews

Title: A Systematic Review of the Psychometric Properties of Quality of Life measures for primary school aged children with Cerebral Palsy

Authors:

Stacey L Carlon (stacey.carlon@gmail.com)
Nora Shields (N.Shields@latrobe.edu.au)
Katherine M Yong (kmy23katherine@hotmail.com)
Rose Gilmore (rose.gilmore@optusnet.com.au)
Leanne Sakzewski (leanne.sakzewski@uqconnect.edu.au)
Roslyn N Boyd (r.boyd@uq.edu.au)

Version: 5 Date: 8 July 2010

Author's response to reviews: see over